



# Article Re-Imagining Trade Policy and Energy Efficiency: Groundbreaking Pathways to Strengthen Environmental Sustainability in South Korea

Dongxue Wang<sup>1</sup> and Yugang He<sup>2,\*</sup>

- <sup>1</sup> Department of International Trade, Woosuk University, Jeonju 55338, Republic of Korea
- <sup>2</sup> Department of Chinese Trade and Commerce, Sejong University, Seoul 05006, Republic of Korea
- \* Correspondence: 1293647581@jbnu.ac.kr

Abstract: This study explores the long-term interplay between trade policy, energy efficiency, and carbon dioxide (CO<sub>2</sub>) emissions in South Korea, using data spanning from 1985 to 2023. By applying the Fourier autoregressive distributed lag (FARDL) model, the analysis reveals that while trade liberalization initially leads to a 0.23% increase in CO<sub>2</sub> emissions for each 1% rise in trade openness driven by the energy demands of industrial expansion-integrating energy efficiency standards within trade agreements helps mitigate these effects over time; this results in a 0.26% reduction in emissions for every 1% improvement in energy efficiency. The study also highlights the dual role of foreign direct investment (FDI), which contributes to a short-term 0.08% rise in emissions but significantly reduces carbon intensity in the long term by facilitating the adoption of cleaner technologies. These findings underscore the importance of innovation and FDI in decoupling economic growth from environmental degradation. The study advocates for the incorporation of energy efficiency measures into trade agreements and the prioritization of green technologies, recommending strategies that could enable South Korea to reduce its CO<sub>2</sub> emissions by up to 40% by 2030. This research positions South Korea as a key actor in achieving global climate goals while maintaining economic competitiveness, offering valuable insights into the balance between sustainable development and industrial growth.

**Keywords:** trade policy; energy efficiency; carbon dioxide emissions; foreign direct investment; environmental sustainability

# 1. Introduction

Addressing climate change has become increasingly urgent, particularly in highly industrialized nations such as South Korea, where balancing rapid economic growth with environmental sustainability poses substantial challenges. As one of the world's major exporters and industrial leaders, South Korea relies heavily on energy-intensive sectors such as petrochemicals, steel, and electronics, contributing to high per capita  $CO_2$  emissions. This reliance on carbon-intensive industries underscores the need for policy interventions that can effectively decouple economic growth from environmental degradation, a challenge that becomes more critical as global climate commitments intensify. Trade policy and energy efficiency are pivotal mechanisms in this context. While South Korea's trade liberalization has facilitated industrial expansion and bolstered economic growth, it has simultaneously intensified CO2 emissions due to the increased energy demands in sectors that are highly carbon-intensive. As documented in the research by Sebri and Ben-Salha [1] and Zafar et al. [2], trade openness, although beneficial for economic growth, often comes at the cost of heightened emissions. However, incorporating energy efficiency standards into trade agreements presents a viable strategy to mitigate these negative externalities. By adopting stringent environmental regulations and promoting energy-efficient technologies, South Korea can lower the carbon intensity of its exports,



Citation: Wang, D.; He, Y. Re-Imagining Trade Policy and Energy Efficiency: Groundbreaking Pathways to Strengthen Environmental Sustainability in South Korea. *Appl. Sci.* **2024**, *14*, 9443. https://doi.org/10.3390/app14209443

Academic Editor: Tymon Zieliński

Received: 26 September 2024 Revised: 11 October 2024 Accepted: 15 October 2024 Published: 16 October 2024



**Copyright:** © 2024 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). aligning trade policies with both domestic and global sustainability goals. Furthermore, the role of green technology innovation is crucial in addressing South Korea's environmental challenges. Recent studies emphasize the potential of renewable energy systems, carbon capture technologies, and advanced energy management innovations to significantly reduce CO<sub>2</sub> emissions (Gielen et al. [3]; Mikulčić et al. [4]). Technological advancements, supported by robust policy frameworks, are vital in accelerating South Korea's transition towards a low-carbon economy. In addition, foreign direct investment (FDI) serves as a critical avenue for emissions reduction, as it facilitates the transfer of cleaner technologies and sustainable practices, contributing to the decoupling of industrial growth from environmental harm. Nasir et al. [5] and Essandoh et al. [6] have highlighted the importance of FDI in promoting the adoption of energy-efficient practices across sectors.

South Korea's commitment to reducing  $CO_2$  emissions by 40% by 2030, compared to 2018 levels, underscores the pressing need for a swift transition to a low-carbon economy. While the Korean Emissions Trading System provides a crucial policy foundation for incentivizing emissions reductions, a more comprehensive integration of environmental principles into trade and industrial strategies is vital to achieve the nation's ambitious climate goals. This study offers critical insights by quantifying the long-term impacts of trade policy, energy efficiency, and foreign direct investment on  $CO_2$  emissions, providing evidence to inform sustainable development pathways for South Korea. Utilizing the advanced Fourier autoregressive distributed lag (FARDL) methodology, which accounts for both short- and long-term effects and adjusts for structural breaks and non-linearities, the research highlights the immediate rise in  $CO_2$  emissions driven by trade liberalization, particularly in energy-intensive industries. Over time, however, the inclusion of energy efficiency standards within trade agreements mitigates these impacts, leading to a net decrease in emissions. The study further emphasizes the significant role of FDI and advancements in green technology in reducing carbon intensity, showcasing the importance of aligning trade policies with environmental sustainability goals. The findings offer robust empirical support for the promotion of energy-efficient technologies and environmentally conscious trade policies as pivotal elements in South Korea's journey toward a low-carbon future. These results contribute to the broader discourse on the interactions between trade, energy policy, and environmental sustainability, providing valuable guidance for policymakers seeking to balance economic growth with climate change mitigation efforts.

This study offers four significant contributions to the literature, particularly in the realm of environmental sustainability within South Korea. First, while existing research has explored the relationship between trade liberalization and CO<sub>2</sub> emissions, this study innovates by quantitatively analyzing the short- and long-term effects of trade policy, energy efficiency, and foreign direct investment using the Fourier autoregressive distributed lag model. This methodological advancement addresses limitations of traditional models, offering a more detailed understanding of the dynamic interactions between these variables, and accounting for structural breaks and non-linear patterns often overlooked in previous studies. Second, while much of the existing scholarship focuses on economic growth and industrialization as primary drivers of emissions, this study introduces energy efficiency as a central factor. The findings reveal that energy efficiency plays a dual role: in the short term, it mitigates the negative externalities associated with trade expansion, while over the long term, it decouples economic growth from carbon emissions. This contribution stands out as earlier studies, such as Rahman et al. [7] and Chikezie Ekwueme et al. [8], have not adequately considered the cumulative impact of energy efficiency on emissions reduction over time. Third, this study broadens the conversation on FDI by highlighting its divergent short- and long-term impacts on emissions in South Korea. Initially, FDI is associated with increased emissions due to higher industrial activity but, over time, it supports the adoption of cleaner, more energy-efficient technologies. This analysis deepens the insights provided by Lee and Zhao [9] and Khan et al. [10], who primarily focus on the economic advantages of FDI without fully addressing its environmental consequences. Lastly, the research offers a comprehensive policy framework that aligns trade policy with environmental

sustainability objectives, recommending the integration of energy efficiency and green technology innovation into South Korea's trade agreements. This forward-thinking policy perspective fills a gap in the literature, as prior research has tended to focus on technological adoption within national borders without sufficiently exploring the global dimensions of trade and sustainability. By underscoring the role of international collaboration, this study sets a new standard for future research in environmental economics and policy development.

The structure of this paper is outlined as follows: Section 2 provides a thorough review of the relevant literature that underpins the research. Section 3 introduces the methodology, including the formulation of variables and model specification. Section 4 presents an in-depth analysis of the empirical results, offering a detailed interpretation of the findings. Finally, Section 5 summarizes the study's conclusions, highlighting key insights and their broader implications for policy and future research directions.

## 2. Literature Review

The interaction between trade policy and environmental sustainability has gained increasing attention in the global conversation on climate change mitigation, particularly in the context of highly industrialized nations such as South Korea. Central to this discourse is the Environmental Kuznets Curve hypothesis, which posits that economic growth initially intensifies environmental degradation before improving once a certain level of income is reached. Recent studies, such as those by Zeeshan et al. [11] and Zhu et al. [12], offer compelling evidence supporting this hypothesis, illustrating that, while trade liberalization promotes economic growth, it also leads to a short-term rise in  $CO_2$  emissions, driven by the energy demands of industrial expansion. Nevertheless, as highlighted by Sajeev and Kaur [13] and Jiang et al. [14], trade openness alone cannot explain the complexities of environmental outcomes. The incorporation of energy efficiency standards and the promotion of green technologies, as noted by Corrocher and Mancusi [15] and Shahzad et al. [16], have become critical components of South Korea's trade agreements, providing a viable pathway to mitigate the environmental costs associated with industrial growth. These interventions are essential, not only for reducing carbon emissions but also for facilitating the transition to a more sustainable growth model. This study extends the existing literature by quantifying the role of energy efficiency in reducing carbon emissions, while also emphasizing the critical function of foreign direct investment in accelerating the adoption of cleaner technologies. Through a detailed examination of these factors, the study sheds light on how FDI can be strategically leveraged to support South Korea's broader sustainability goals, offering valuable insights for policymakers aiming to balance economic development with environmental stewardship.

Green technology innovation and improvements in energy efficiency are essential elements in the long-term decoupling of economic growth from environmental degradation. These components are integral to addressing the global challenge of reducing CO<sub>2</sub> emissions while maintaining robust economic performance. Razmjoo et al. [17] and Madejski et al. [18] underscore the transformative potential of renewable energy systems and carbon capture technologies in achieving significant emissions reductions. In the case of South Korea, energy efficiency has emerged as a crucial factor in moderating the carbon intensity of its export-oriented economy. Oryani et al. [19] and Ding and Lee [20] have demonstrated that the incorporation of energy efficiency standards within trade agreements has led to a substantial decline in energy consumption per unit of output in key industrial sectors, particularly in carbon-intensive industries such as steel and petrochemicals. This shift has been pivotal in reducing the environmental footprint of these sectors, which have traditionally been major contributors to the nation's  $CO_2$  emissions. However, much of the existing literature, including research by Sonnenschein and Mundaca [21] and Ha and Byrne [22], has tended to focus on the immediate impacts of energy efficiency improvements, often neglecting their cumulative, long-term effects. This study addresses this oversight by providing empirical evidence that demonstrates how sustained advancements in energy

efficiency and green technology can significantly reduce emissions over time. The results highlight that, while the short-term gains are measurable, it is the long-term commitment to energy innovation that drives meaningful environmental progress. Such findings underscore the importance of a forward-looking policy framework that prioritizes continuous investments in energy efficiency and innovation, especially in industrialized nations like South Korea. Moreover, foreign direct investment plays a critical role in facilitating this transition toward sustainability. Dos Santos Gaspar et al. [23] and Apostu and Gigauri [24] emphasize the capacity of FDI to act as a conduit for the transfer of cleaner, more efficient technologies across borders. By enabling access to advanced technological innovations, FDI supports the adoption of energy-efficient practices in industries that are otherwise slow to innovate due to financial or technical constraints. This study further demonstrates that FDI not only promotes industrial growth but also reinforces sustainable development by enhancing the capacity of domestic industries to integrate cleaner technologies. The synergistic effects of FDI, energy efficiency, and green technology create a comprehensive pathway toward achieving the dual objectives of economic growth and environmental sustainability, offering a model for policymakers to optimize trade and investment strategies in alignment with global climate goals.

Although energy efficiency and green technologies have demonstrated clear positive impacts on environmental sustainability, the role of foreign direct investment in this context remains a subject of ongoing debate within the literature. FDI, while a catalyst for industrial growth, often leads to an initial increase in  $CO_2$  emissions, as industrial expansion drives higher energy consumption. This phenomenon is well documented by Nepal et al. [25] and Tariq et al. [26], who note that the influx of foreign capital typically stimulates energyintensive production in the early stages of industrial development. However, the long-term effects of FDI tell a more nuanced story. Over time, FDI plays a pivotal role in facilitating the transfer of advanced energy-efficient technologies and practices, as evidenced by Huan and Qamruzzaman [27] and Amara et al. [28]. These studies suggest that, while FDI may initially contribute to environmental degradation, its ultimate impact is a net reduction in emissions as industries adopt cleaner technologies and more efficient energy management systems. This research extends the existing literature by examining the divergent temporal impacts of FDI, providing fresh insights into how foreign investment contributes to emissions reduction. The analysis reveals that FDI not only drives industrial growth but also supports South Korea's long-term environmental goals by enabling the adoption of advanced energy management practices. By facilitating the integration of green technologies, FDI becomes a critical tool for decoupling economic growth from environmental harm. This finding aligns with the broader understanding of FDI's dual role, which initially exacerbates emissions but ultimately mitigates them through technological innovation and transfer. Additionally, the results underscore the necessity of aligning FDI flows with South Korea's overarching sustainability objectives. Policies that incentivize investments in green technologies are vital for ensuring that FDI serves as a driver of both economic and environmental progress. This perspective echoes the recommendations of Kim and Seok [29] and Ghorbal et al. [30], who emphasize the importance of strategically channeling FDI into sectors that prioritize sustainability. Thus, the study not only contributes to the theoretical understanding of FDI's environmental implications but also offers actionable policy guidance for aligning foreign investments with national and global climate objectives.

In comparison to previous research, this study makes notable contributions by providing a more refined understanding of the dynamic relationship between trade policy, energy efficiency, and carbon emissions. Earlier studies, including those by Park and Hong [31], Salman et al. [32], and Khan et al. [33], have extensively examined the roles of economic growth and urbanization in contributing to rising emissions. However, these studies have not sufficiently addressed the mitigating effects that energy efficiency can exert over time. This research fills that gap by employing a robust methodological framework, specifically the Fourier autoregressive distributed lag model, to capture both short- and long-term impacts of trade policy and energy efficiency on carbon emissions in South Korea. The results present compelling evidence that, while trade liberalization initially leads to higher emissions, the integration of energy efficiency standards into trade agreements can progressively reverse this upward trend.

## 3. Variables and Model

## 3.1. Variables

Independent variable: Carbon dioxide emissions serve as a critical indicator of South Korea's environmental sustainability, closely linked to the country's industrial composition, energy consumption patterns, and policy frameworks. As a highly industrialized nation with significant reliance on fossil fuels, South Korea has experienced elevated levels of greenhouse gas emissions, positioning  $CO_2$  emissions as a prominent measure of its environmental impact. Studies by He [34], Li and He [35], and Wang et al. [36] emphasize the strong relationship between  $CO_2$  emissions, energy use, and economic growth in South Korea. The nation's rapid urbanization and industrial expansion have driven emissions upward, although recent investments in green technologies and energy-efficient practices have begun to temper this trajectory. Despite these advancements, South Korea remains among the world's leading  $CO_2$  emitters, highlighting the urgency of transitioning to a low-carbon economy to meet international climate objectives. The government's commitment to reducing CO<sub>2</sub> emissions by 40% by 2030, relative to 2018 levels, underscores the centrality of  $CO_2$  as a metric for both environmental performance and policy direction. Initiatives such as the Korean Emissions Trading System reinforce the pivotal role of  $CO_2$ emissions in guiding South Korea's sustainable development strategy (Adebayo et al. [37]). Consequently, CO<sub>2</sub> emissions are not only a key indicator of environmental sustainability but also serve as a fundamental variable in assessing the broader economic-environmental nexus in academic research. Their inclusion in this analysis reflects the country's ongoing efforts to balance robust economic growth with stringent environmental responsibilities.

Core variable: Trade policy and energy efficiency are key determinants of South Korea's carbon dioxide emissions, particularly due to the country's position as a heavily industrialized and export-driven economy. Recent authoritative studies highlight that South Korea's liberal trade policies, while driving economic growth, have also led to elevated CO<sub>2</sub> emissions. This is largely due to the energy-intensive nature of key industries such as steel and petrochemicals (Koc and Bulus [38]; Adebayo et al. [39]; He [40]). The expansion of exports, facilitated by trade liberalization, has resulted in increased energy consumption and, consequently, higher emissions. However, evidence also suggests that the incorporation of energy efficiency measures within trade policies can counteract this upward trend in emissions. For example, Moon and Min [41] and Nam and Jin [42] demonstrate that advancements in energy efficiency, spurred by domestic regulations and international environmental agreements, have reduced the energy intensity of South Korea's exports. This shift has facilitated a partial decoupling of economic growth from CO<sub>2</sub> emissions, marking an important step towards sustainability. Further, the adoption of cutting-edge technologies and the implementation of stringent energy efficiency standards promoted through various trade agreements-have enabled South Korean industries to lower  $CO_2$  emissions per unit of output (Lee [43]; Raihan [44]; Cho et al. [45]). Thus, while trade expansion has traditionally been linked to higher emissions, these findings underscore the positive impact of integrating energy efficiency into trade frameworks. Moving forward, the alignment of trade policies with robust energy efficiency standards will be crucial to South Korea's pursuit of long-term carbon reduction goals and sustainable economic growth.

Control variable: Economic growth, urbanization, industrialization, foreign direct investment, and green technology innovation are pivotal drivers of South Korea's CO<sub>2</sub> emissions. Historically, economic growth has been linked to higher energy consumption, particularly in energy-intensive manufacturing sectors, leading to increased CO<sub>2</sub> emissions (Kim [46]; Song et al. [47]). The process of urbanization further intensifies energy demand, as expanding infrastructure, housing, and transportation needs contribute signifi-

cantly to the country's emission profile (Shafique et al. [48]). Similarly, industrialization, a cornerstone of South Korea's economic strategy, exacerbates emissions, with sectors like petrochemicals and steel production heavily reliant on fossil fuels (Yun and Jeong [49]; Xu et al. [50]). On the other hand, FDI exerts a more nuanced influence. While FDI often boosts industrial output and energy use, it also facilitates the transfer of cleaner, more energy-efficient technologies, which can mitigate  $CO_2$  emissions over time (Kim [46]). Finally, green technology innovation plays a crucial role in reducing emissions. Innovations in renewable energy, energy efficiency, and carbon capture are progressively decoupling economic growth from emissions, indicating a shift toward a more sustainable future (Jung et al. [51]). Together, these factors form a complex landscape that influences both the short-term rise in emissions and long-term mitigation strategies in South Korea. To provide a clearer understanding of the variables analyzed in this paper, Table 1 summarizes the essential characteristics of each variable.

Table 1. Description of variables.

Variable	Form	Definition	Source
Carbon dioxide emissions	CO <sub>2</sub>	Metric tons per capita in log	World Bank
Trade policy	tp	Trade policy index	FRED Economic Database
energy efficiency	ee	GDP/energy consumption	World Bank
Economic growth	eg	GDP (constant 2015 USD, and unit: billion USD) in log	World Bank
Urbanization	ur	Share of the urban population in the total population	World Bank
Industrialization	in	Ratio of industry to GDP	World Bank
Foreign direct investment	fdi	Foreign direct investment, net inflows (% of GDP)	World Bank
Green technology innovation	gt	Values for patents (% of total) in environment-related technologies	Organization for Economic Cooperation and Development

# 3.2. Model

In analyzing the long-term interaction between trade policy, energy efficiency, and carbon dioxide emissions in South Korea, it is critical to account for both the economic and environmental dimensions that have underpinned the country's development trajectory. South Korea's export-oriented industrialization, heavily dependent on energy-intensive sectors such as manufacturing and petrochemicals, has historically contributed to rising  $CO_2$  emissions (Hong et al. [52]). Nevertheless, recent empirical research indicates that advancements in energy efficiency, catalyzed by domestic regulatory frameworks and international trade agreements, have the potential to alleviate the environmental pressures associated with industrial growth (Kim and Tromp [53]). The integration of energy efficiency measures within trade policies is particularly important for decoupling economic growth from environmental degradation. This suggests that increased trade openness, when aligned with robust energy efficiency standards, can support long-term reductions in  $CO_2$  emissions (Hille and Lambernd [54]; Wenlong et al. [55]). Building on these insights, this paper introduces a long-term empirical model to quantify the effects of trade policy and energy efficiency on  $CO_2$  emissions, offering a comprehensive framework for understanding the pathways to sustainable economic growth. The baseline model is outlined as follows:

$$CO_{2,t} = a_0 + a_1 t p_t + a_2 e e_t + a_3 e g_t + a_4 u r_t + a_5 i n_t + a_6 f d i_t + a_7 g t_t + \epsilon_t.$$
(1)

In Equation (1), t represents the time variable, denoting the year under consideration.  $a_0$  refers to the constant term, which captures the baseline level of the dependent variable when all other explanatory variables are zero. The terms  $[a_1, a_7]$  correspond to the coefficients to be estimated, reflecting the magnitude and direction of the relationships between the independent variables and the dependent variable. Finally,  $\epsilon$  denotes the white noise error term, which accounts for the random variability not explained by the model, ensuring that the residuals are normally distributed with a mean of zero. This formulation allows for the precise estimation of the impact of key variables on the outcome while maintaining statistical rigor.

In addition, this study employs the innovative Fourier autoregressive distributed lag methodology to achieve its research objectives. The following section provides a detailed discussion of this approach. Traditional methods for testing cointegration, such as the Engle and Granger [56] two-step procedure and the Johansen and Juselius [57] test, are widely recognized in the literature. However, these techniques are constrained by certain limitations, particularly the requirement for all variables to be integrated at the same order. To address these issues, Pesaran et al. [58] introduced the autoregressive distributed lag (ARDL) model, which offers greater flexibility by allowing variables of mixed orders of integration (i.e., I(0) or I(1)). The ARDL model also addresses challenges such as serial correlation and endogeneity, providing more reliable and robust estimates. Moreover, it distinguishes between short- and long-run dynamics, enabling the formulation of policies that are tailored to different time horizons. The ARDL framework, therefore, offers a comprehensive approach to understanding both short- and long-term relationships between variables, as demonstrated in the following formal model specification.

$$\Delta CO_{2,t} = b_0 + \sum_{i=0}^{n_1} b_{1i} \Delta CO_{2,t-i} + \sum_{i=0}^{n_2} b_{2i} \Delta tp_{t-i} + \sum_{i=0}^{n_3} b_{3i} \Delta ee_{t-i} + \sum_{i=0}^{n_4} b_{4i} \Delta eg_{t-i} + \sum_{i=0}^{n_5} b_{5i} \Delta ur_{t-i} + \sum_{i=0}^{n_6} b_{6i} \Delta in_{t-i} + \sum_{i=0}^{n_7} b_{7i} \Delta fd_{t-i} + \sum_{i=0}^{n_8} b_{8i} \Delta gt_{t-i} + b_9 CO_{2,t-1} + b_{10} tp_{t-1} + b_{11} ee_{t-1} + b_{12} eg_{t-1} + b_{13} ur_{t-1} + b_{14} in_{t-1} + b_{15} fd_{t-1} + b_{16} gt_{t-1} + ect_{t-1} + \epsilon_t.$$

Equation (2) illustrates the ARDL model with an unrestricted intercept and no trend. In this formulation,  $b_0$  represents the intercept, while the coefficients  $[b_{1i}, b_{8i}]$  correspond to short-run estimates and  $[b_9, b_{16}]$  reflect the long-run estimates. The terms [n1, n8] denote the lag orders, and  $\epsilon_t$  represents the disturbance term. To assess cointegration, two standard tests are applied within the ARDL framework: firstly, an F-test (or bounds test) on the longrun coefficients, testing the null hypothesis that  $b_9 = b_{10} = b_{11} = b_{12} = b_{13} = b_{14} = b_{15} = b_{15}$  $b_{16} = 0$ , and secondly, a *t*-test on the lagged dependent variable, specifically testing  $b_9 = 0$ . The null hypothesis  $H_0$  in both cases posits no cointegrating relationship, whereas the alternative hypothesis  $H_1$  asserts the presence of cointegration. Cointegration is confirmed if the F- and t-test statistics exceed the critical values for the upper bound, corresponding to an I(1) series. However, one limitation of the ARDL approach is its inability to account for structural breaks, which can lead to biased results (Enders and Lee [59]; Gil-Alana and Yaya [60]; Cai and Omay [61]). To address this issue, the Fourier transformation is incorporated into the ARDL model, enhancing its ability to handle structural breaks without requiring prior knowledge of the timing, frequency, or form of the breaks. Unlike structural break dummies, the Fourier transformation efficiently captures these breaks while maintaining strong power and size properties due to its reduced number of parameters. The model incorporating the Fourier transformation, known as FARDL, is formulated as follows:

$$\Delta CO_{2,t} = b_0 + \sum_{i=0}^{n1} b_{1i} \Delta CO_{2,t-i} + \sum_{i=0}^{n2} b_{2i} \Delta tp_{t-i} + \sum_{i=0}^{n3} b_{3i} \Delta ee_{t-i} + \sum_{i=0}^{n4} b_{4i} \Delta eg_{t-i} + \sum_{i=0}^{n5} b_{5i} \Delta ur_{t-i} + \sum_{i=0}^{n6} b_{6i} \Delta in_{t-i} + \sum_{i=0}^{n7} b_{7i} \Delta fdi_{t-i} + \sum_{i=0}^{n8} b_{8i} \Delta gt_{t-i} + b_9 CO_{2,t-1} + b_{10} tp_{t-1} + b_{11} ee_{t-1} + b_{12} eg_{t-1} + b_{13} ur_{t-1} + b_{14} in_{t-1} + b_{15} fdi_{t-1} + b_{16} gt_{t-1} + F_1 sin\left(\frac{2\pi kt}{T}\right) + F_2 cos\left(\frac{2\pi kt}{T}\right) + ect_{t-1} + \epsilon_t.$$

Equation (3) introduces two key components of the Fourier transformation:  $F_1$  and  $F_2$ , which represent the amplitude and displacement, respectively. In this context, the term  $\pi$  is the mathematical constant (approximately 3.14), while k denotes the frequency parameter associated with the Fourier series. The variable t captures the time trend, and T refers to the total sample size. The Fourier transformation enhances the model by allowing for smooth, periodic shifts in the data, which can capture underlying cyclical patterns and

structural changes without relying on predefined breakpoints. This approach is particularly advantageous in econometric models where abrupt shifts in the underlying data-generating process might bias traditional estimates. The inclusion of Fourier terms in ARDL models has been shown to improve model flexibility and fit by capturing non-linearities that might otherwise be overlooked (Wu et al. [62]; Alper et al. [63]; Syed et al. [64]). Moreover, this method allows the identification of structural breaks without the need for prior assumptions about their timing, thus offering a more nuanced approach to understanding long-run relationships in the presence of such breaks.

#### 4. Results and Discussion

## 4.1. Basic Statistical Analysis

This section presents the empirical results of the analysis. In time series econometrics, testing for unit roots or stationarity is a critical preliminary step to avoid spurious regression results and to ensure the selection of the appropriate estimation method. Accordingly, before proceeding with the estimation of the FARDL model, unit root tests were conducted to determine the order of integration for each variable. Traditional unit root tests, such as the augmented Dickey–Fuller (ADF) test and the Phillips–Perron (PP) test, are commonly employed in this context. However, these conventional methods do not account for structural breaks, which can lead to misleading conclusions in the presence of such breaks. To address this limitation and enhance the robustness of the analysis, the Fourier-ADF and Fourier-LM tests were utilized. These tests incorporate Fourier transformations, which effectively capture structural breaks by identifying smooth shifts in the data without requiring prior knowledge of the breakpoints. The use of Fourier-based unit root tests allows for a more reliable assessment of stationarity, accommodating potential non-linearities and discontinuities in the data. As a result, this approach yields more robust and accurate empirical findings. The results from the Fourier-ADF and Fourier-LM tests, presented in Table 2, provide a comprehensive overview of the stationarity properties of the variables under investigation, ensuring a solid foundation for subsequent model estimation and policy analysis.

Variable ——	Fourier-	Fourier-ADF Test		Fourier-LM Test	
	I(0)	I(1)	I(0)	I(1)	
CO <sub>2</sub>	0.137	-5.796 ***	-4.489 **	-6.357 ***	
tp	-2.652	-6.771 ***	-1.635	-6.856 ***	
ee	-1.972	-6.132 ***	-1.838	-6.538 ***	
eg	-0.723	-5.856 ***	-4.502 **	-5.924 ***	
ur	-1.745	-5.003 ***	-1.136	-5.399 ***	
in	-1.593	-6.782 ***	-4.131 **	-6.898 ***	
fdi	-2.474	-5.515 ***	-1.098	-6.146 ***	
gt	-1.041	-5.619 ***	-1.283	-6.395 ***	

Table 2. Results of unit root test.

Note: \*\* 5% significance level. \*\*\* 1% significance level.

The null hypothesis  $H_0$  for both the Fourier-LM and Fourier-ADF tests posits the presence of a unit root, while the alternative hypothesis  $H_1$  suggests stationarity. As shown in Table 2, the results indicate that  $H_0$  cannot be rejected at the level I(0) for either test, signifying that the dataset exhibits non-stationarity at levels. However, upon differencing the data,  $H_0$  is rejected at I(1), confirming that the variables become stationary after the first difference. This finding implies that the series is integrated of order one, I(1), and thus suitable for further analysis using the FARDL framework. The decision to employ the FARDL approach in this study is particularly justified by its ability to account for structural breaks, a feature that traditional cointegration methods often overlook. As highlighted in the methodology, the FARDL method outperforms conventional approaches by incorporating Fourier terms, which capture smooth structural shifts without prior

knowledge of their timing. This flexibility is crucial for improving the robustness of the empirical results, especially in the context of time series data with potential breaks. The results of the FARDL bounds testing, as summarized in Table 3, provide further evidence of the long-run relationships between the variables under study, offering insights into the dynamic interactions that govern the model. This methodology enhances the reliability of the analysis, ensuring that both short- and long-term impacts are appropriately captured, even in the presence of structural changes.

Table 3. Results of cointegration test.

Method	Statistical Value	Lower Critical Value	Upper Critical Value
<i>t</i> -statistical test	-7.742 ***	-2.58	-4.23
F-statistical test	21.218 ***	3.07	4.44

Note: \*\*\* 1% significance level.

As presented in Table 3, the results from both the F-test and *t*-test indicate that the respective test statistics exceed the critical values for the upper bound. Consequently, the null hypothesis of no cointegration is decisively rejected. This suggests the existence of a stable long-run relationship among the variables under consideration in this study. The rejection of the null hypothesis confirms that the variables are cointegrated, signifying that despite short-term fluctuations, they tend to move together in the long run. Following this, the analysis proceeds to examine the long-run and short-run dynamics as estimated by the FARDL model. The FARDL approach allows for a detailed decomposition of the long-term equilibrium relationships and short-term adjustments, enabling a more nuanced understanding of the interactions between the variables. The subsequent section presents the estimated coefficients for both the long-run and short-run relationships, offering insights into the extent and direction of the effects, as well as the speed of adjustment toward long-run equilibrium. These findings provide a robust basis for policy recommendations, as they illuminate the differential impacts of the considered variables over various time horizons.

#### 4.2. Long- and Short-Run Effect Test Analysis

To extend the analysis from the cointegration results to a more granular examination of the long- and short-run dynamics, it is critical to assess the implications of the FARDL model across different time horizons. The results of the cointegration tests have confirmed a robust long-term relationship between the key variables, indicating that despite temporary deviations, these variables converge towards a stable long-run equilibrium. In this context, the subsequent examination of short-run dynamics will provide valuable insights into the immediate, transient effects of trade policy, energy efficiency, and other influencing factors on  $CO_2$  emissions. Conversely, the long-run estimates will capture the enduring, equilibrium adjustments that unfold over time. Table 4 offers a detailed presentation of these findings, facilitating a comprehensive understanding of how the determinants impact  $CO_2$  emissions over both short and long durations. These results highlight not only the elasticity of  $CO_2$  emissions in the short term but also the broader, sustained effects of the key variables in shaping South Korea's long-term environmental outcomes.

As demonstrated in Table 4, short-term effects of South Korea's trade policy are marked by increased industrial activity, particularly in energy-intensive sectors such as manufacturing and petrochemicals, which leads to a rise in  $CO_2$  emissions. This outcome is largely attributable to the rapid expansion of production capacities and heightened energy consumption facilitated by trade liberalization (Li et al. [65]; Chen et al. [66]). The opening of markets and greater export activity, in the absence of strict environmental regulations, tends to elevate emissions, as highlighted by Hu et al. [67] and Wang et al. [68]. These studies observe that the initial phases of trade liberalization contribute significantly to  $CO_2$  emissions as industries scale up to meet growing international demand. In contrast, over the long term, the relationship between trade policy and  $CO_2$  emissions shifts as energy efficiency measures and technological advancements are gradually integrated. The

adoption of environmental standards within trade agreements, alongside innovations in green technologies, plays a critical role in reducing the carbon intensity of production. Nathaniel et al. [69] and Işık et al. [70] provide evidence supporting this, demonstrating that trade policies, when aligned with environmental regulations, can reduce the energy intensity of exports and result in declining  $CO_2$  emissions over time. This aligns with the findings of the FARDL model in this study, which shows a negative long-term impact of trade policy on  $CO_2$  emissions. The primary distinction between this study and prior research lies in the treatment of the transitional phase. While earlier studies acknowledge the initial rise in emissions due to trade liberalization, this study offers a more detailed perspective by emphasizing the long-term mitigating effects of energy efficiency. Both studies converge on the conclusion that trade policies can have differing impacts across time horizons, with short-term increases in emissions ultimately offset by long-term reductions driven by technological progress and regulatory improvements.

Variable	Long-Run Effect	Variable	Short-Run Effect
tp	-0.093 *** (-6.239)	Δtp	0.229 *** (5.714)
ee	-0.259 *** (-3.689)	Δee	-0.147 *** (-3.916)
eg	0.359 *** (8.524)	Δeg	0.533 *** (8.866)
ur	0.199 ** (2.205)	Δur	0.258 *** (2.834)
in	0.306 ** (2.129)	Δin	0.447 *** (3.738)
fdi	-0.108 ** (-2.273)	Δfdi	0.075 ** (2.215)
gt	-0.117 *** (-3.228)	$\Delta \mathrm{gt}$	-0.032 *** (-3.662)
Diagnostics test			
Statistical test	Value	Statistical test	Value
ect	-0.181 ***	Ramsey RESET test	0.152
$F_1$	0.052 ***	ARCH test	0.292
F <sub>2</sub>	0.079 ***	Jarque–Bera test	0.381
LM test	0.217	CUSUM and CUSUM of squares test	Stable

Table 4. Results of long- and short-run effect test analysis.

Note: \*\*\* 1% significance level. \*\* 5% significance level. *t*-statistics in the parentheses.

In the short term, energy efficiency has a negative effect on South Korea's CO<sub>2</sub> emissions, primarily by reducing the energy intensity of production processes, particularly within energy-demanding industries such as manufacturing. The immediate reductions in emissions stem from the adoption of more efficient technologies, improved energy management systems, and the optimization of industrial operations. These early gains, although relatively moderate due to the gradual integration of new practices and technologies, contribute to lower  $CO_2$  emissions even as industrial output continues to grow. This is consistent with findings from Steren et al. [71] and Hossain et al. [72], who emphasize the short-term advantages of energy efficiency measures, although their full potential may only be realized as these innovations become more widely implemented. In the long run, the impact of energy efficiency on  $CO_2$  emissions is even more substantial. Sustained investment in energy-efficient technologies, along with advancements in renewable energy, leads to a more pronounced decoupling of economic growth from carbon emissions. The compounding effects of technological innovation, policy support, and broader adoption of green practices significantly enhance the reduction of emissions over time. This is corroborated by Misila et al. [73] and Nam and Jin [42], who highlight the importance of energy efficiency as a key driver of long-term emissions reduction, particularly when combined

with advancements in clean technologies and supportive regulatory frameworks. The primary distinction between the findings of this study and those of previous literature lies in the more granular differentiation between short- and long-term effects. While existing studies acknowledge the overall role of energy efficiency in reducing emissions, this study offers a more comprehensive perspective by clearly delineating the time horizons over which these impacts manifest. Nonetheless, the conclusion remains consistent across studies: the long-term benefits of energy efficiency are significantly greater than the short-term reductions, especially when underpinned by continuous technological innovation and policy interventions.

In the short term, South Korea's rapid economic growth, urbanization, and industrialization contribute significantly to increasing CO<sub>2</sub> emissions, primarily due to the country's dependence on energy-intensive industries and urban infrastructure expansion. Economic growth in particular drives a higher demand for energy across critical sectors, such as manufacturing, transportation, and construction, leading to increased consumption of fossil fuels and, consequently, higher emissions. Similarly, urbanization, characterized by the growth of cities and rising demand for services like housing and transportation, exacerbates energy use, further elevating emissions. Industrialization, especially in sectors like steel, petrochemicals, and electronics, plays a crucial role in driving emissions due to the high carbon intensity of these industries (Park et al. [74]; Tan et al. [75]). Over the long term, the positive relationship between CO<sub>2</sub> emissions, economic growth, urbanization, and industrialization gradually weakens. This is due to several factors, including structural shifts toward a service-based economy, improvements in energy efficiency, and an increasing reliance on renewable energy. As the economy evolves, technological advancements and policy measures, such as the Korean Emissions Trading System, mitigate the environmental impact of these factors. Although economic activities still contribute to emissions, the long-term effects are less pronounced compared to the short-term impacts. This observation is supported by Brockway et al. [76] and Lamb et al. [77], who find that industrial expansion initially leads to higher emissions, but over time, technological and structural changes reduce these effects.

Foreign direct investment follows a similar trajectory. In the short term, FDI typically results in greater industrial activity and emissions, as foreign capital often flows into energy-intensive sectors that are central to South Korea's economy. The increased production fueled by FDI elevates energy consumption, thereby contributing to higher CO2 emissions. However, in the long run, FDI has the potential to reduce emissions by facilitating the transfer of cleaner, more energy-efficient technologies. Studies by Hille et al. [78] and Saqib and Dincă [79] highlight the critical role FDI plays in introducing advanced technologies that help reduce the carbon intensity of industrial processes. As these technologies are integrated into South Korea's industrial base, the long-term impact of FDI on CO<sub>2</sub> emissions becomes negative, with emissions decreasing as energy efficiency improves. Green technology innovation exerts a negative influence on  $CO_2$  emissions in both the short and long term. In the short term, the introduction of green technologies, such as renewable energy systems, energy-efficient industrial processes, and carbon capture technologies, begins to lower the carbon intensity of production and consumption. These early effects, although initially modest due to the time required for widespread adoption, contribute to emissions reductions. Over the long term, however, the impact of green technology innovation becomes increasingly significant. As innovation accelerates and these technologies are more widely adopted across industries, the reductions in CO2 emissions become more substantial. This is consistent with studies by Lee and Woo [80] and Mo [81], which emphasize that sustained innovation in green technologies, particularly when supported by favorable policy frameworks, is crucial to long-term emissions reductions.

When comparing these findings with the existing literature, both convergence and divergence can be observed. The short-term positive relationship between economic growth, urbanization, and industrialization with CO<sub>2</sub> emissions is consistent with studies by Pata [82], Odugbesan and Rjoub [83], and Jiang et al. [84], which underscore the immedi-

ate environmental costs associated with rapid industrial expansion and urban development. These studies, like the present analysis, emphasize that the short-term benefits of economic growth in terms of emissions reductions are constrained by the significant energy demands that accompany early-stage industrialization and urbanization. However, this study distinguishes itself by offering a more nuanced perspective on the varying effects of these factors over time, particularly regarding green technology innovation. While prior studies, such as those by Bulus and Koc [85], acknowledge the importance of green technology in reducing long-term emissions, this study extends the discussion by highlighting the compounding effects of continuous technological innovation and integration. It demonstrates that, while the initial impact of green technology on emissions may be limited, its long-term effects are far more substantial, particularly as green technologies become more affordable and widely adopted. Furthermore, the long-term negative impact of FDI on CO<sub>2</sub> emissions, as found in this study, aligns with the findings of Wang and Huang [86], all of which emphasize the importance of FDI in facilitating the transfer of cleaner technologies. However, this study contributes additional depth to the discussion by clearly distinguishing between the short-term and long-term impacts of FDI, noting that while FDI initially increases emissions, its long-term effects can be mitigated through the widespread adoption of energy-efficient technologies. This difference highlights the importance of time horizons when evaluating the environmental impacts of FDI.

In conclusion, this study aligns with much of the existing literature, particularly in recognizing the divergent impacts of economic growth, urbanization, industrialization, FDI, and green technology innovation on  $CO_2$  emissions over different time periods. The analysis presented here, however, offers a more detailed and comprehensive understanding of how these factors evolve over time, providing valuable insights into both short- and long-term pathways to emissions reductions in South Korea. As the country continues to balance economic growth with environmental sustainability, these findings underscore the critical importance of sustained technological innovation, targeted policy interventions, and strategic use of FDI to achieve long-term carbon reduction goals.

## 4.3. Robustness Test Analysis

To validate the robustness of the empirical findings, this analysis applies three econometric techniques: fully modified ordinary least squares (FMOLS), dynamic ordinary least squares (DOLS), and canonical cointegration regression (CCR). These methods are designed to address potential econometric issues such as endogeneity and serial correlation, thereby ensuring the reliability and accuracy of the long-run coefficient estimates. FMOLS is particularly effective in adjusting for serial correlation and endogeneity, making it well-suited for dealing with small sample sizes and time series that are integrated. DOLS further enhances the robustness by incorporating leads and lags of differenced variables, which corrects for potential biases in ordinary least squares (OLS) and yields consistent estimates in the context of cointegrated relationships. Similarly, CCR transforms the original data before estimation to tackle endogeneity and serial correlation, providing additional robustness to the long-term estimates. By employing these three methods, the study ensures that the estimated long-run relationships, particularly those involving trade policy, energy efficiency, and  $CO_2$  emissions, are rigorously tested and validated. The results obtained from the FMOLS, DOLS, and CCR estimations, as presented in Table 5, confirm the consistency and robustness of the key findings, offering further assurance in the credibility and reliability of the conclusions drawn from the analysis.

As evidenced in Table 5, all estimated coefficients are statistically significant at the 1% level, underscoring the robustness of the results across the FMOLS, DOLS, and CCR models. Notably, the negative coefficients for trade policy and energy efficiency across all models indicate that enhancements in both areas contribute to a reduction in  $CO_2$  emissions, thus improving environmental quality and promoting sustainability in South Korea. These findings are consistent with the baseline results and further confirm that policies designed to liberalize trade, when coupled with improvements in energy efficiency, have a substantial

positive impact on reducing emissions. In the case of trade policy, these results suggest that the strategic integration of environmental regulations within trade agreements can play a pivotal role in mitigating the environmental footprint of industrial activities, particularly in sectors that are energy-intensive. Similarly, improvements in energy efficiency contribute directly to emissions reduction by lowering the energy consumption per unit of output across industries. This is particularly relevant for South Korea, where economic growth has traditionally been tied to energy-intensive manufacturing sectors. These findings resonate with previous studies that highlight the dual benefits of trade liberalization and energy efficiency in reducing environmental degradation (Woo et al. [87]; Zhang [88]; Kim and Lin [89]). The consistency of the results across FMOLS, DOLS, and CCR further enhances the credibility of the conclusions, as these methodologies are specifically designed to correct for issues such as endogeneity and serial correlation, ensuring that the estimated relationships reflect true long-term effects. Consequently, the negative coefficients not only corroborate the baseline findings but also provide additional evidence that policies aimed at improving energy efficiency and trade regulation can be key drivers of long-term environmental sustainability in South Korea. Thus, the results offer robust support for policy frameworks that emphasize sustainable trade practices and technological innovation in energy efficiency as essential components of national and global efforts to reduce CO<sub>2</sub> emissions.

Table 5. Results of robustness test analysis.

Variable	FMOLS	DOLS	CCR
tp	-0.126 ***	-0.053 ***	-0.077 ***
	(-7.233)	(-7.199)	(-6.887)
ee	-0.292 ***	-0.109 ***	-0.136 ***
	(-4.113)	(-4.074)	(-4.254)
CV	Yes	Yes	Yes

Note: \*\*\* 1% significance level. *t*-statistics in the parentheses. cv control variable.

## 5. Conclusions

This study underscores the pivotal influence of trade policy, energy efficiency, and various economic factors in determining South Korea's carbon emissions trajectory. In both the short and long term, improvements in energy efficiency and the adoption of environmentally sustainable trade practices emerge as essential strategies for reducing CO<sub>2</sub> emissions. The analysis demonstrates that, while economic growth, urbanization, and industrialization initially drive up emissions, the deployment of advanced technologies, foreign direct investment, and green innovations significantly offset these effects in the long run. South Korea's robust commitment to environmental sustainability, reinforced by its progressive policies and technological advancements, provides a solid framework for achieving its ambitious carbon reduction objectives. As the nation continues to balance the dual imperatives of economic growth and environmental stewardship, the findings offer compelling support for the adoption of long-term strategies that integrate sustainability into development. By advancing energy efficiency standards and accelerating the adoption of green technology, South Korea is well positioned to become a global leader in the transition toward a low-carbon economy, making a substantial contribution to international efforts to mitigate climate change.

Drawing on the results of this study, several key policy implications and strategic solutions are proposed to advance environmental sustainability while maintaining economic growth. First, it is essential to align trade policies with stringent environmental regulations to reduce the carbon emissions associated with industrial exports. Incorporating energy efficiency standards into trade agreements can significantly lower the carbon intensity of South Korea's manufacturing exports. The Korean Emissions Trading System offers a foundational mechanism to support these efforts, incentivizing industries to adopt cleaner and more sustainable production practices. Second, the government should focus on increasing investments in energy efficiency, particularly within energy-intensive sectors

such as petrochemicals, steel, and electronics. Implementing policies that encourage the adoption of state-of-the-art technologies to reduce energy consumption per unit of output is crucial. Offering subsidies or tax incentives to industries meeting enhanced energy efficiency benchmarks can further stimulate innovation and contribute to the overall reduction of CO<sub>2</sub> emissions. Third, promoting innovation in green technologies, including renewable energy and carbon capture systems, should be a key priority. Government funding, research grants, and partnerships with private sector actors are essential for accelerating the development and widespread implementation of these technologies. Such efforts are critical to decoupling economic growth from environmental degradation and achieving substantial long-term reductions in CO<sub>2</sub> emissions. Finally, South Korea should strategically target foreign direct investment that facilitates the transfer of cleaner and more energy-efficient technologies into the domestic market. FDI policies should prioritize partnerships with foreign firms that emphasize sustainability, ensuring that foreign capital not only enhances industrial capacity but also contributes to emissions reductions over time. These policy recommendations offer a comprehensive framework for achieving significant reductions in CO<sub>2</sub> emissions while supporting South Korea's broader economic objectives, ensuring that the country remains competitive globally while pursuing its environmental sustainability goals.

This study presents several limitations that highlight potential directions for future research. First, the analysis relies on national-level data, potentially masking regional disparities in CO<sub>2</sub> emissions across South Korea's industrial and urban areas. Future studies could evaluate region-specific emissions and policy impacts, offering a more detailed understanding of geographic variability in environmental outcomes. Second, while this research employs advanced econometric methods to explore the relationship between trade policy, energy efficiency, and  $CO_2$  emissions, it does not fully consider the effects of global economic fluctuations or geopolitical events on trade dynamics. Future research could integrate the impacts of international market volatility and geopolitical tensions, providing a more comprehensive assessment of South Korea's environmental sustainability in a global context. Third, the study focuses primarily on energy efficiency and trade policy but does not sufficiently examine the role of emerging technologies, such as artificial intelligence and blockchain, in reducing emissions. Further research could assess the potential of these advanced technologies to enhance industrial practices and accelerate emissions reductions. Lastly, although the study addresses the long-term impact of foreign direct investment on CO<sub>2</sub> emissions, further examination of sector-specific FDI and its environmental implications could provide deeper insights. Future research could investigate how directing FDI toward green technology sectors may expedite South Korea's transition to a low-carbon economy, contributing to both national sustainability goals and global climate commitments. These avenues for future inquiry would enrich the understanding of South Korea's environmental challenges and guide more targeted, effective policy interventions.

**Author Contributions:** Conceptualization, Y.H.; methodology, Y.H.; software, D.W.; validation, D.W.; formal analysis, D.W.; investigation, D.W.; resources, D.W.; data curation, D.W.; writing—original draft preparation, D.W.; writing—review and editing, Y.H.; visualization, D.W. All authors have read and agreed to the published version of the manuscript.

Funding: This research received no external funding.

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

**Data Availability Statement:** The data presented in this study are available from the authors upon request.

Conflicts of Interest: The authors declare no conflicts of interest.

## References

- Sebri, M.; Ben-Salha, O. On the Causal Dynamics between Economic Growth, Renewable Energy Consumption, CO<sub>2</sub> Emissions and Trade Openness: Fresh Evidence from BRICS Countries. *Renew. Sustain. Energy Rev.* 2014, 39, 14–23. [CrossRef]
- Zafar, M.W.; Mirza, F.M.; Zaidi, S.A.H.; Hou, F. The Nexus of Renewable and Nonrenewable Energy Consumption, Trade Openness, and CO<sub>2</sub> Emissions in the Framework of EKC: Evidence from Emerging Economies. *Environ. Sci. Pollut. Res.* 2019, 26, 15162–15173. [CrossRef] [PubMed]
- 3. Gielen, D.; Boshell, F.; Saygin, D.; Bazilian, M.D.; Wagner, N.; Gorini, R. The Role of Renewable Energy in the Global Energy Transformation. *Energy Strategy Rev.* **2019**, *24*, 38–50. [CrossRef]
- Mikulčić, H.; Skov, I.R.; Dominković, D.F.; Alwi, S.R.W.; Manan, Z.A.; Tan, R.; Duić, N.; Mohamad, S.N.H.; Wang, X. Flexible Carbon Capture and Utilization Technologies in Future Energy Systems and the Utilization Pathways of Captured CO<sub>2</sub>. *Renew. Sustain. Energy Rev.* 2019, 114, 109338. [CrossRef]
- Nasir, M.A.; Huynh, T.L.D.; Tram, H.T.X. Role of Financial Development, Economic Growth & Foreign Direct Investment in Driving Climate Change: A Case of Emerging ASEAN. J. Environ. Manag. 2019, 242, 131–141.
- Essandoh, O.K.; Islam, M.; Kakinaka, M. Linking International Trade and Foreign Direct Investment to CO<sub>2</sub> Emissions: Any Differences between Developed and Developing Countries? *Sci. Total Environ.* 2020, 712, 136437. [CrossRef]
- Rahman, M.M.; Nepal, R.; Alam, K. Impacts of Human Capital, Exports, Economic Growth and Energy Consumption on CO<sub>2</sub> Emissions of a Cross-Sectionally Dependent Panel: Evidence from the Newly Industrialized Countries (NICs). *Environ. Sci. Policy* 2021, 121, 24–36. [CrossRef]
- 8. Chikezie Ekwueme, D.; Lasisi, T.T.; Eluwole, K.K. Environmental Sustainability in Asian Countries: Understanding the Criticality of Economic Growth, Industrialization, Tourism Import, and Energy Use. *Energy Environ.* 2023, 34, 1592–1618. [CrossRef]
- 9. Lee, C.-C.; Zhao, Y.-N. Heterogeneity Analysis of Factors Influencing CO<sub>2</sub> Emissions: The Role of Human Capital, Urbanization, and FDI. *Renew. Sustain. Energy Rev.* 2023, 185, 113644. [CrossRef]
- Khan, M.; Rana, A.T.; Ghardallou, W. FDI and CO<sub>2</sub> Emissions in Developing Countries: The Role of Human Capital. *Nat. Hazards* 2023, 117, 1125–1155. [CrossRef]
- Zeeshan, M.; Han, J.; Rehman, A.; Ullah, I.; Afridi, F.E.A.; Fareed, Z. Comparative Analysis of Trade Liberalization, CO<sub>2</sub> Emissions, Energy Consumption and Economic Growth in Southeast Asian and Latin American Regions: A Structural Equation Modeling Approach. *Front. Environ. Sci.* 2022, 10, 854590. [CrossRef]
- Zhu, K.; Ali, A.; Zhang, T.; Zada, M. An Empirical Investigation of the Impact of Energy Consumption, Globalization and Natural Resources on Ecological Footprint and Economic Growth, Evidence from China, Japan, South Korea and China Taiwan. *Energy Environ.* 2024, 0958305X241251421. [CrossRef]
- 13. Sajeev, A.; Kaur, S. Environmental Sustainability, Trade and Economic Growth in India: Implications for Public Policy. *Int. Trade Politics Dev.* **2020**, *4*, 141–160. [CrossRef]
- 14. Jiang, S.; Mentel, G.; Shahzadi, I.; Jebli, M.B.; Iqbal, N. Renewable Energy, Trade Diversification and Environmental Footprints: Evidence for Asia-Pacific Economic Cooperation (APEC). *Renew. Energy* **2022**, *187*, 874–886. [CrossRef]
- 15. Corrocher, N.; Mancusi, M.L. International Collaborations in Green Energy Technologies: What Is the Role of Distance in Environmental Policy Stringency? *Energy Policy* **2021**, *156*, 112470. [CrossRef]
- Shahzad, U.; Radulescu, M.; Rahim, S.; Isik, C.; Yousaf, Z.; Ionescu, S.A. Do Environment-Related Policy Instruments and Technologies Facilitate Renewable Energy Generation? Exploring the Contextual Evidence from Developed Economies. *Energies* 2021, 14, 690. [CrossRef]
- Razmjoo, A.; Kaigutha, L.G.; Rad, M.V.; Marzband, M.; Davarpanah, A.; Denai, M. A Technical Analysis Investigating Energy Sustainability Utilizing Reliable Renewable Energy Sources to Reduce CO<sub>2</sub> Emissions in a High Potential Area. *Renew. Energy* 2021, 164, 46–57. [CrossRef]
- 18. Madejski, P.; Chmiel, K.; Subramanian, N.; Kuś, T. Methods and Techniques for CO<sub>2</sub> Capture: Review of Potential Solutions and Applications in Modern Energy Technologies. *Energies* **2022**, *15*, 887. [CrossRef]
- 19. Oryani, B.; Moridian, A.; Rezania, S.; Vasseghian, Y.; Bagheri, M.; Shahzad, K. Asymmetric Impacts of Economic Uncertainties and Energy Consumption on the Ecological Footprint: Implications apropos Structural Transformation in South Korea. *Fuel* **2022**, 322, 124180. [CrossRef]
- 20. Ding, H.; Lee, W. ESG and Financial Performance of China Firms: The Mediating Role of Export Share and Moderating Role of Carbon Intensity. *Sustainability* **2024**, *16*, 5042. [CrossRef]
- Sonnenschein, J.; Mundaca, L. Decarbonization under Green Growth Strategies? The Case of South Korea. J. Clean. Prod. 2016, 123, 180–193. [CrossRef]
- Ha, Y.; Byrne, J. The Rise and Fall of Green Growth: Korea's Energy Sector Experiment and Its Lessons for Sustainable Energy Policy. WIREs Energy Environ. 2019, 8, e335. [CrossRef]
- 23. dos Santos Gaspar, J.; Marques, A.C.; Fuinhas, J.A. The Traditional Energy-Growth Nexus: A Comparison between Sustainable Development and Economic Growth Approaches. *Ecol. Indic.* 2017, *75*, 286–296. [CrossRef]
- 24. Apostu, S.-A.; Gigauri, I. Sustainable Development and Entrepreneurship in Emerging Countries: Are Sustainable Development and Entrepreneurship Reciprocally Reinforcing? *J. Entrep. Manag. Innov.* **2023**, *19*, 41–77. [CrossRef]
- Nepal, R.; Paija, N.; Tyagi, B.; Harvie, C. Energy Security, Economic Growth and Environmental Sustainability in India: Does FDI and Trade Openness Play a Role? J. Environ. Manag. 2021, 281, 111886. [CrossRef]

- Tariq, G.; Sun, H.; Ali, I.; Pasha, A.A.; Khan, M.S.; Rahman, M.M.; Mohamed, A.; Shah, Q. Influence of Green Technology, Green Energy Consumption, Energy Efficiency, Trade, Economic Development and FDI on Climate Change in South Asia. *Sci. Rep.* 2022, 12, 16376. [CrossRef]
- 27. Huan, Y.; Qamruzzaman, M. Innovation-Led FDI Sustainability: Clarifying the Nexus between Financial Innovation, Technological Innovation, Environmental Innovation, and FDI in the BRIC Nations. *Sustainability* **2022**, *14*, 15732. [CrossRef]
- Amara, D.B.; Qiao, J.; Zada, M. How to Reconcile the Climate Change Issue with Economic Growth? Spatial Dual Mediating Effects of Carbon Emissions and Foreign Investment. J. Clean. Prod. 2023, 411, 137285. [CrossRef]
- 29. Kim, S.-E.; Seok, J.H. The Impact of Foreign Direct Investment on CO<sub>2</sub> Emissions Considering Economic Development: Evidence from South Korea. J. Int. Trade Econ. Dev. **2023**, 32, 537–552. [CrossRef]
- Ghorbal, S.; Soltani, L.; Ben Youssef, S. Patents, Fossil Fuels, Foreign Direct Investment, and Carbon Dioxide Emissions in South Korea. *Environ. Dev. Sustain.* 2022, 26, 109–125. [CrossRef]
- Park, J.; Hong, T. Analysis of South Korea's Economic Growth, Carbon Dioxide Emission, and Energy Consumption Using the Markov Switching Model. *Renew. Sustain. Energy Rev.* 2013, 18, 543–551. [CrossRef]
- 32. Salman, M.; Long, X.; Dauda, L.; Mensah, C.N. The Impact of Institutional Quality on Economic Growth and Carbon Emissions: Evidence from Indonesia, South Korea and Thailand. *J. Clean. Prod.* **2019**, 241, 118331. [CrossRef]
- 33. Khan, K.; Su, C.-W.; Tao, R.; Hao, L.-N. Urbanization and Carbon Emission: Causality Evidence from the New Industrialized Economies. *Environ. Dev. Sustain.* 2020, 22, 7193–7213. [CrossRef]
- 34. He, Y. Investigating the Routes Toward Environmental Sustainability: Fresh Insights from Korea. *Sustainability* **2022**, *15*, 602. [CrossRef]
- Li, Y.; He, Y. Unraveling Korea's Energy Challenge: The Consequences of Carbon Dioxide Emissions and Energy Use on Economic Sustainability. Sustainability 2024, 16, 2074. [CrossRef]
- 36. Wang, L.; He, Y.; Wu, R. The Green Engine of Growth: Assessing the Influence of Renewable Energy Consumption and Environmental Policy on China's Economic Sustainability. *Sustainability* **2024**, *16*, 3120. [CrossRef]
- 37. Adebayo, T.S.; Coelho, M.F.; Onbaşıoğlu, D.Ç.; Rjoub, H.; Mata, M.N.; Carvalho, P.V.; Rita, J.X.; Adeshola, I. Modeling the Dynamic Linkage between Renewable Energy Consumption, Globalization, and Environmental Degradation in South Korea: Does Technological Innovation Matter? *Energies* 2021, 14, 4265. [CrossRef]
- Koc, S.; Bulus, G.C. Testing Validity of the EKC Hypothesis in South Korea: Role of Renewable Energy and Trade Openness. Environ. Sci. Pollut. Res. 2020, 27, 29043–29054. [CrossRef]
- Adebayo, T.S.; Awosusi, A.A.; Kirikkaleli, D.; Akinsola, G.D.; Mwamba, M.N. Can CO<sub>2</sub> Emissions and Energy Consumption Determine the Economic Performance of South Korea? A Time Series Analysis. *Environ. Sci. Pollut. Res.* 2021, 28, 38969–38984. [CrossRef]
- He, Y. Unraveling the Interplay between Food Security, Agriculture, Trade Policy, and Energy Consumption: An Environmental Sustainability Insight. *Energy Environ.* 2023, 0958305X231195604. [CrossRef]
- Moon, H.; Min, D. A DEA Approach for Evaluating the Relationship between Energy Efficiency and Financial Performance for Energy-Intensive Firms in Korea. J. Clean. Prod. 2020, 255, 120283. [CrossRef]
- 42. Nam, E.; Jin, T. Mitigating Carbon Emissions by Energy Transition, Energy Efficiency, and Electrification: Difference between Regulation Indicators and Empirical Data. J. Clean. Prod. 2021, 300, 126962. [CrossRef]
- Lee, H. Is Carbon Neutrality Feasible for Korean Manufacturing Firms?: The CO<sub>2</sub> Emissions Performance of the Metafrontier Malmquist–Luenberger Index. J. Environ. Manag. 2021, 297, 113235. [CrossRef] [PubMed]
- 44. Raihan, A. Nexus between Greenhouse Gas Emissions and Its Determinants: The Role of Renewable Energy and Technological Innovations towards Green Development in South Korea. *Innov. Green Dev.* **2023**, *2*, 100066. [CrossRef]
- 45. Cho, B.-K.; Jung, H.; Chung, J.-B.; Song, C.-K. Implications of the Carbon Border Adjustment Mechanism on South Korean Industries: Challenges and Policy Recommendations. J. Clean. Prod. 2024, 444, 141278. [CrossRef]
- 46. Kim, S. The Effects of Foreign Direct Investment, Economic Growth, Industrial Structure, Renewable and Nuclear Energy, and Urbanization on Korean Greenhouse Gas Emissions. *Sustainability* **2020**, *12*, 1625. [CrossRef]
- Song, M.-J.; Seo, Y.-J.; Lee, H.-Y. The Dynamic Relationship between Industrialization, Urbanization, CO<sub>2</sub> Emissions, and Transportation Modes in Korea: Empirical Evidence from Maritime and Air Transport. *Transportation* 2023, 50, 2111–2137. [CrossRef]
- Shafique, M.; Azam, A.; Rafiq, M.; Luo, X. Evaluating the Relationship between Freight Transport, Economic Prosperity, Urbanization, and CO<sub>2</sub> Emissions: Evidence from Hong Kong, Singapore, and South Korea. *Sustainability* 2020, 12, 10664. [CrossRef]
- 49. Yun, J.; Jeong, S. Contributions of Economic Growth, Terrestrial Sinks, and Atmospheric Transport to the Increasing Atmospheric CO2 Concentrations over the Korean Peninsula. *Carbon Balance Manag.* **2021**, *16*, 22. [CrossRef]
- 50. Xu, G.; Wang, Y.; Rehman, H. The Future Trajectory of Carbon Emissions in the Process of Carbon Neutrality in South Korea. *J. Environ. Manag.* **2023**, *345*, 118588. [CrossRef]
- 51. Jung, S.; Kim, H.; Kang, Y.; Jeong, E. Analysis of Korea's Green Technology Policy and Investment Trends for the Realization of Carbon Neutrality: Focusing on CCUS Technology. *Processes* **2022**, *10*, 501. [CrossRef]
- 52. Hong, J.H.; Kim, J.; Son, W.; Shin, H.; Kim, N.; Lee, W.K.; Kim, J. Long-Term Energy Strategy Scenarios for South Korea: Transition to a Sustainable Energy System. *Energy Policy* **2019**, *127*, 425–437. [CrossRef]

- 53. Kim, T.-J.; Tromp, N. Analysis of Carbon Emissions Embodied in South Korea's International Trade: Production-Based and Consumption-Based Perspectives. *J. Clean. Prod.* 2021, 320, 128839. [CrossRef]
- 54. Hille, E.; Lambernd, B. The Role of Innovation in Reducing South Korea's Energy Intensity: Regional-Data Evidence on Various Energy Carriers. *J. Environ. Manag.* 2020, 262, 110293. [CrossRef]
- Wenlong, Z.; Tien, N.H.; Sibghatullah, A.; Asih, D.; Soelton, M.; Ramli, Y. Impact of Energy Efficiency, Technology Innovation, Institutional Quality, and Trade Openness on Greenhouse Gas Emissions in Ten Asian Economies. *Environ. Sci. Pollut. Res.* 2022, 30, 43024–43039. [CrossRef] [PubMed]
- Engle, R.F.; Granger, C.W. Co-Integration and Error Correction: Representation, Estimation, and Testing. *Econom. J. Econom. Soc.* 1987, 55, 251–276. [CrossRef]
- Johansen, S.; Juselius, K. Maximum Likelihood Estimation and Inference on Cointegration—With Appucations to the Demand for Money. Oxf. Bull. Econ. Stat. 1990, 52, 169–210. [CrossRef]
- Pesaran, M.H.; Shin, Y.; Smith, R.J. Bounds Testing Approaches to the Analysis of Level Relationships. J. Appl. Econom. 2001, 16, 289–326. [CrossRef]
- Enders, W.; Lee, J. A Unit Root Test Using a Fourier Series to Approximate Smooth Breaks. Oxf. Bull. Econ. Stat. 2012, 74, 574–599. [CrossRef]
- Gil-Alana, L.A.; Yaya, O.S. Testing Fractional Unit Roots with Non-Linear Smooth Break Approximations Using Fourier Functions. J. Appl. Stat. 2021, 48, 2542–2559. [CrossRef]
- 61. Cai, Y.; Omay, T. Using Double Frequency in Fourier Dickey–Fuller Unit Root Test. Comput. Econ. 2022, 59, 445–470. [CrossRef]
- Wu, C.-F.; Chang, T.; Wu, T.-P.; Leng, K.; Lin, M.-C.; Huang, S.-C. Impact of Globalization on the Environment in Major CO<sub>2</sub>-Emitting Countries: Evidence Using Bootstrap ARDL with a Fourier Function. *Front. Public Health* 2022, 10, 907403. [CrossRef] [PubMed]
- 63. Alper, A.E.; Alper, F.O.; Ozayturk, G.; Mike, F. Testing the Long-Run Impact of Economic Growth, Energy Consumption, and Globalization on Ecological Footprint: New Evidence from Fourier Bootstrap ARDL and Fourier Bootstrap Toda–Yamamoto Test Results. *Environ. Sci. Pollut. Res.* 2022, *30*, 42873–42888. [CrossRef] [PubMed]
- 64. Syed, Q.R.; Apergis, N.; Goh, S.K. The Dynamic Relationship between Climate Policy Uncertainty and Renewable Energy in the US: Applying the Novel Fourier Augmented Autoregressive Distributed Lags Approach. *Energy* **2023**, *275*, 127383. [CrossRef]
- 65. Li, M.; Ahmad, M.; Fareed, Z.; Hassan, T.; Kirikkaleli, D. Role of Trade Openness, Export Diversification, and Renewable Electricity Output in Realizing Carbon Neutrality Dream of China. *J. Environ. Manag.* **2021**, 297, 113419. [CrossRef]
- Chen, S.; Zhang, H.; Wang, S. Trade Openness, Economic Growth, and Energy Intensity in China. *Technol. Forecast. Soc. Chang.* 2022, 179, 121608. [CrossRef]
- 67. Hu, X.; Pollitt, H.; Pirie, J.; Mercure, J.-F.; Liu, J.; Meng, J.; Tao, S. The Impacts of the Trade Liberalization of Environmental Goods on Power System and CO<sub>2</sub> Emissions. *Energy Policy* **2020**, *140*, 111173. [CrossRef]
- 68. Wang, Q.; Zhang, F.; Li, R. Free Trade and Carbon Emissions Revisited: The Asymmetric Impacts of Trade Diversification and Trade Openness. *Sustain. Dev.* **2024**, *32*, 876–901. [CrossRef]
- 69. Nathaniel, S.P.; Murshed, M.; Bassim, M. The Nexus between Economic Growth, Energy Use, International Trade and Ecological Footprints: The Role of Environmental Regulations in N11 Countries. *Energy Ecol. Environ.* **2021**, *6*, 496–512. [CrossRef]
- Işık, C.; Ongan, S.; Ozdemir, D.; Jabeen, G.; Sharif, A.; Alvarado, R.; Amin, A.; Rehman, A. Renewable Energy, Climate Policy Uncertainty, Industrial Production, Domestic Exports/Re-Exports, and CO<sub>2</sub> Emissions in the USA: A SVAR Approach. *Gondwana Res.* 2024, 127, 156–164. [CrossRef]
- 71. Steren, A.; Rubin, O.D.; Rosenzweig, S. Energy-Efficiency Policies Targeting Consumers May Not Save Energy in the Long Run: A Rebound Effect That Cannot Be Ignored. *Energy Res. Soc. Sci.* **2022**, *90*, 102600. [CrossRef]
- Hossain, M.E.; Rej, S.; Saha, S.M.; Onwe, J.C.; Nwulu, N.; Bekun, F.V.; Taha, A. Can Energy Efficiency Help in Achieving Carbon-Neutrality Pledges? A Developing Country Perspective Using Dynamic ARDL Simulations. *Sustainability* 2022, 14, 7537. [CrossRef]
- Misila, P.; Winyuchakrit, P.; Limmeechokchai, B. Thailand's Long-Term GHG Emission Reduction in 2050: The Achievement of Renewable Energy and Energy Efficiency beyond the NDC. *Heliyon* 2020, 6, e05720. [CrossRef] [PubMed]
- 74. Park, D.U.; Park, H.; Kim, Y.; Baek, S.; Lee, D.; Jang, Y.; Jung, T.; Rodriguez, F.S. *The Role of Science, Technology and Innovation Policies in the Industrialization of Developing Countries*; STEPI: Sejong-si, Republic of Korea, 2021.
- Tan, X.; Wang, R.; Choi, Y.; Lee, H. Does Korea's Carbon Emissions Trading Scheme Enhance Efficiency for Sustainable Energy and Utilities? *Util. Policy* 2024, 88, 101752. [CrossRef]
- 76. Brockway, P.E.; Sorrell, S.; Semieniuk, G.; Heun, M.K.; Court, V. Energy Efficiency and Economy-Wide Rebound Effects: A Review of the Evidence and Its Implications. *Renew. Sustain. Energy Rev.* **2021**, *141*, 110781. [CrossRef]
- 77. Lamb, W.F.; Wiedmann, T.; Pongratz, J.; Andrew, R.; Crippa, M.; Olivier, J.G.; Wiedenhofer, D.; Mattioli, G.; Al Khourdajie, A.; House, J. A Review of Trends and Drivers of Greenhouse Gas Emissions by Sector from 1990 to 2018. *Environ. Res. Lett.* 2021, 16, 073005. [CrossRef]
- 78. Hille, E.; Shahbaz, M.; Moosa, I. The Impact of FDI on Regional Air Pollution in the Republic of Korea: A Way Ahead to Achieve the Green Growth Strategy? *Energy Econ.* 2019, *81*, 308–326. [CrossRef]
- 79. Saqib, N.; Dincă, G. Exploring the Asymmetric Impact of Economic Complexity, FDI, and Green Technology on Carbon Emissions: Policy Stringency for Clean-Energy Investing Countries. *Geosci. Front.* **2024**, *15*, 101671. [CrossRef]

- 80. Lee, J.-H.; Woo, J. Green New Deal Policy of South Korea: Policy Innovation for a Sustainability Transition. *Sustainability* **2020**, *12*, 10191. [CrossRef]
- Mo, J.Y. Technological Innovation and Its Impact on Carbon Emissions: Evidence from Korea Manufacturing Firms Participating Emission Trading Scheme. *Technol. Anal. Strateg. Manag.* 2022, 34, 47–57. [CrossRef]
- Pata, U.K. The Effect of Urbanization and Industrialization on Carbon Emissions in Turkey: Evidence from ARDL Bounds Testing Procedure. *Environ. Sci. Pollut. Res.* 2018, 25, 7740–7747. [CrossRef] [PubMed]
- 83. Odugbesan, J.A.; Rjoub, H. Relationship among Economic Growth, Energy Consumption, CO<sub>2</sub> Emission, and Urbanization: Evidence from MINT Countries. *SAGE Open* **2020**, *10*, 215824402091464. [CrossRef]
- 84. Jiang, J.; Zhu, S.; Wang, W. Carbon Emissions, Economic Growth, Urbanization, and Foreign Trade in China: Empirical Evidence from ARDL Models. *Sustainability* **2022**, *14*, 9396. [CrossRef]
- 85. Bulus, G.C.; Koc, S. The Effects of FDI and Government Expenditures on Environmental Pollution in Korea: The Pollution Haven Hypothesis Revisited. *Environ. Sci. Pollut. Res.* 2021, *28*, 38238–38253. [CrossRef] [PubMed]
- Wang, Y.; Huang, Y. Impact of Foreign Direct Investment on the Carbon Dioxide Emissions of East Asian Countries Based on a Panel ARDL Method. *Front. Environ. Sci.* 2022, 10, 937837. [CrossRef]
- Woo, C.; Chung, Y.; Chun, D.; Seo, H.; Hong, S. The Static and Dynamic Environmental Efficiency of Renewable Energy: A Malmquist Index Analysis of OECD Countries. *Renew. Sustain. Energy Rev.* 2015, 47, 367–376. [CrossRef]
- 88. Zhang, S. Is Trade Openness Good for Environment in South Korea? The Role of Non-Fossil Electricity Consumption. *Environ. Sci. Pollut. Res.* 2018, 25, 9510–9522. [CrossRef]
- 89. Kim, D.-H.; Lin, S.-C. Trade Openness and Environmental Policy Stringency: Quantile Evidence. *Sustainability* **2022**, *14*, 3590. [CrossRef]

**Disclaimer/Publisher's Note:** The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.